Author’s response to reviews

Title: De-medicalization of birth by reducing the use of oxytocin for augmentation among first time mothers - a prospective intervention study

Authors:
Lise Gaudernack (lisegaudernack@yahoo.no)
Kathrine Frøslie (k.f.froslie@medisin.uio.no)
Trond Michelsen (trmi1@ous-hf.no)
Nanna Voldner (nvoldner@hotmail.com)
Mirjam Lukasse (mirjam.lukasse@hioa.no)

Version: 2 Date: 03 Dec 2017

Author’s response to reviews:

Dear Editor

We are grateful for your comments and have implemented the changes:
- Line 299: the word correct has been changed to "appropriate"
- Table 3 CS: caesarean section has been removed
- Regarding figure 1 we have made the figure simpler and easier to understand for the reader.

Regarding the payment, the invoice can be sent to:

Oslo and Akershus University College of Applied Sciences
P.O. Box 364 Alnabru
N-0614 OSLO

Oslo and Akershus University College of Applied Sciences (HIOA) prefer to receive invoices via HIOA’s electronic invoice address: NO997058925.
Please mark the invoice: Reference code: 2210, Project: 191036

Yours sincerely

Lise C Gaudernack